PS							
Machine Coordinator last week			PSR & TT2 (Gianfranco EN-ACE), SWY (Marcel, Oliver BE-OP)				
Machine Coord	inator this we	eek P	PSR & TT2 (Gianfranco EN-ACE), SWY (Marcel, Oliver BE-OP)				
Beam Scheduled							
East Area	Yes/No	nTOF	Yes/No	AD	Yes/No	SPS	Yes/No
Beam Availability by Destination (AFT)							
AD	%	EA N	%	EA T8	%	EA T9	%
nTOF	%	SPS %					
Facility Status							
Summary	PSR & TT2: YETS SWY: hardware commissioning has started 12/02.						
This week	Monday: IST for PS main units (under responsibility of EN-ACE) Tuesday: DSO test for SWY, PSR and TT2 (under responsibility of EN-ACE) Since Wednesday: HWC for SWY. Short accesses for KFA28 and F16.BTV135 were required. Thursday: scheduled accesses for heat run, polarity check of three aux. magnets (exchanged during YETS) and audio/visual patrol of all aux.magnets. Acc-testing tested on testable test equipment with HWC-test cycle Friday: Completion of HWC checklist for SWY						
Next week	Monday 17/02: - MTG intervention from 9:00 to 12:00 - Pulsing SMH26 - Access for SY-BI, intervention on PU37 amplifier Non conformity of fire wall below PS-East area will be fixed. Re-patrol of PS sector 5 will be required - Re-patrol TT2 sector 5, lost after HSE-RP intervention Saturday 22/02: (in practice Monday 24/02) PSR and TT2 handover from EN-ACE to OP.						
Intervention Request							
Yes / No	Duration		Pref	erred date	e/time		
Reason							
Impact							